
\section{Measurability by inspection}
%
\label{sec:obviously}

Definitions broadly like the one in \cref{thm:polynomial-Vitali} are fairy common in mathematics, yet it is also a relatively universal experience that, in practice, one does not need to worry about measurability of functions one has `actually written down.' %
To what extent can we explain this phenomenon?

One suspect in \cref{thm:polynomial-Vitali} is the fact that polynomials are not in general uniformly continuous. Given any uniformly continuous function $g(x,y)$, $f(x) = \sup_y g(x,y)$ is also uniformly continuous and therefore measurable. This is a specific case of a general phenomenon, which is that often ostensibly projective sets or functions are actually Borel. (We'll see a more subtle instance of this in \cref{sec:o-min}.) For instance, often `quantification' (e.g., suprema, infima, intersections, and unions) over an uncountable family can be replaced with quantification over some dense countable subfamily by some monotonicity or continuity condition.






%

\subsection{Measurability of analytic sets}

Roughly speaking,
the strongest\footnote{There are of course always stronger results, such as Solovay's result that `provably $\bDelta^1_2$' sets are Lebesgue measurable \cite[Ex.~14.4]{Kanamori2003}.} \ZFC-provable general theorem regarding measurability is \Lusin's result that analytic sets are always universally measurable.%


%
  %
  %
  %
  %


\begin{defn}
  A subset of a Polish space $X$ is \emph{universally measurable} if it is measurable with regards to the completion of any $\sigma$-finite Borel measure on $X$. A function $f : X \to \olR$ is \emph{universally measurable} if $f^{-1}((r,\infty])$ is universally measurable for every $r \in \olR$.
\end{defn}

Note that the set of universally measurable subsets of a Polish space is always a $\sigma$-algebra, since it is an intersection of $\sigma$-algebras.


\begin{fact}[\Lusin]\label{fact:analytic-univ-meas}
  Any $\bSigma^1_1$ (i.e., analytic) set is universally measurable. %
\end{fact}

Since analytic sets are precisely the images of Borel sets under Borel maps, this class is fairly broad. Furthermore, although complements of analytic sets are not in general analytic, the pointclass of analytic sets is closed under countable unions and intersections (\cref{fact:Bool}).

For the purpose of using \cref{fact:analytic-univ-meas} to show that certain classes of functions are universally measurable, we'll need the following definition and lemma. 

\begin{defn}\label{defn:semi-sigma}
  For any Polish space $X$, a function $f : X \to \olR$ is \emph{(lower) semi-$\bSigma^1_1$} if for every $r \in \olR$, $f^{-1}((r,\infty])$ is $\bSigma^1_1$.
\end{defn}

Unfortunately the term \emph{semianalytic} is both already in use and will appear later in this paper, so we have opted for the visually clunky but unambiguous term in \cref{defn:semi-sigma}. Strictly speaking, the notion of \emph{upper semi-$\bSigma^1_1$} functions clearly also makes sense, but to avoid an overabundance of words we will only use the term `semi-$\bSigma^1_1$' and this will always refer to the notion in \cref{defn:semi-sigma}.

%

\begin{lem}$ $\label{lem:semi-sigma}
  \begin{enumerate}
  \item\label{Borel-char} A function $f$ is Borel if and only if $f$ and $-f$ are both semi-$\bSigma^1_1$. In particular, Borel functions are semi-$\bSigma^1_1$.
  \item\label{univ-meas} Any semi-$\bSigma^1_1$ function is universally measurable.
  \item\label{closed-suplevel} $f$ is semi-$\bSigma^1_1$ if and only if for any $r \in \olR$, $f^{-1}([r,\infty])$ is $\bSigma^1_1$.
    %
  \item\label{countable-quant} If $f: X \times Y \to \olR$ is semi-$\bSigma^1_1$, then $g(x) = \sup_{a \in A}f(x,a)$ and $h(x) = \inf_{a \in A}f(x,a)$ are semi-$\bSigma^1_1$ for any countable $A \subseteq Y$.
  \item\label{uncountable-quant} If $f: X \times Y \to \olR$ is semi-$\bSigma^1_1$, then $\ell(x) = \sup_{y \in Y}f(x,y)$ is semi-$\bSigma^1_1$.
  \end{enumerate}
\end{lem}
\begin{proof}
  \ref{Borel-char} follows from \ref{closed-suplevel} and \cref{fact:Suslin}. \ref{univ-meas} follows immediately from \cref{fact:analytic-univ-meas}. \ref{closed-suplevel} follows from $f^{-1}([r,\infty]) = \bigcap_{s < r,~s \in \Qb}f^{-1}((r,\infty])$, $f^{-1}((r,\infty]) = \bigcup_{s > r,~s \in \Qb}f^{-1}([r,\infty])$ and \cref{fact:Bool}.
  %

  For \ref{countable-quant}, first note that
  \(
    g^{-1}((r,\infty]) = \bigcup_{a \in A}\{ x \in X : \langle x,a \rangle \in f^{-1}((r,\infty])).
  \)
  Since $X \times \{a\}$ is closed in $X \times Y$ for each $a \in A$, the terms in the union are $\bSigma^1_1$, so we have that $g$ is semi-$\bSigma^1_1$. For $h$, we similarly have that $h^{-1}([r,\infty]) = \bigcap_{a \in A}\{x \in X : \langle x,a \rangle \in g^{-1}([r,\infty])\}$, implying that each $h^{-1}([r,\infty])$ is $\bSigma^1_1$ and therefore $h$ is semi-$\bSigma^1_1$ by \ref{closed-suplevel}.

  Finally, for \ref{uncountable-quant}, note that $\ell^{-1}((r,\infty]) = \pi(f^{-1}((r,\infty])))$, where $\pi : X \times Y \to X$ is the projection. Since projections of $\bSigma^1_1$ sets are $\bSigma^1_1$, we have that $\ell$ is semi-$\bSigma^1_1$.
\end{proof}


\subsection{An extra quantifier from \texorpdfstring{$\sigma$}{σ}-compactness}
\label{sec:extra-quant-sigma-compact}



Another suspect in \cref{thm:polynomial-Vitali} is alternation of quantifiers (i.e., alternation of suprema and infima). We can show that $4$ alternating blocks of quantifiers are necessary for something like it to work.

Recall that a topological space is \emph{$\sigma$-compact} if it is a countable union of compact sets. Also recall that a function $g : X \to \olR$ is \emph{lower semi-continuous} if the open superlevel sets $g^{-1}((r,\infty])$ are open for each $r \in \olR$. Note that continuous functions in particular are lower semi-continuous.
%

\begin{prop}\label{prop:three-quant-insufficient}
  Fix a $\sigma$-compact Polish space $Y$ and an arbitrary $\sigma$-compact topological space $Z$. Fix an arbitrary set $W$ (which we will think of as a discrete topological space). Let $g : Y \times Z \times W \to \olR$ be a lower semi-continuous function. Then the function $f : Y \to \olR$ defined by
  \[
    f(y) = \inf_{z \in Z}\sup_{w \in W}g(y,z,w)
  \]
  is Borel. 
  %
  %
\end{prop}
\begin{proof}
  The function $f_0(y,z) = \sup_{w \in W}g(y,z,w)$ is lower semi-continuous, since it is the supremum of a family of lower semi-continuous functions. Therefore the closed sublevel sets $f_0^{-1}([-\infty,r]) \coloneq \{(y,z) : f_0(y,z) \leq r\}$ are closed. Since $Y$ and $Z$ are $\sigma$-compact, their product is as well and so each of the sets $f_0^{-1}([-\infty,r])$ is $\sigma$-compact.

  Now $f(y) = \inf_{z \in Z}f_0(y,z)$. We have that $f^{-1}([-\infty,r)) \coloneq \{y : f(y) < r\}$ is equal to $\bigcup_{s < r}\pi_Z(f_0^{-1}([-\infty,s]))$ (where $\pi_Z : Y \times Z \to Z$ is the canonical projection map). By monotonicity, we also have that $f^{-1}([-\infty,r)) = \bigcup_{s < r,~s \in \Qb}\pi_Z(f_0^{-1}([-\infty,s]))$. For each $s$, we have that since $f_0^{-1}([-\infty,s])$ is $\sigma$-compact, $\pi_Z(g_0^{-1}([-\infty,s]))$ is as well and is hence also $F_\sigma$ (since $Y$ is Hausdorff). Therefore each $f^{-1}([-\infty,r))$ is also $F_\sigma$, since it is a countable union of $F_\sigma$ sets.
%
  %
  %
\end{proof}

\begin{cor}\label{cor:three-quant-insufficient-1}
  If $X$, $Y$, $Z$, and $W$, are Polish spaces (with $Y$ and $Z$ $\sigma$-compact) and $g : X \times Y \times Z \times W \to \olR$ is lower semi-continuous, then
  \[
    f(x) = \sup_{y \in Y}\inf_{z \in Z}\sup_{w \in W}g(x,y,z,w)
  \]
   is semi-$\bSigma^1_1$ and therefore universally measurable.
\end{cor}

One thing to note is that \cref{prop:three-quant-insufficient} does need $\sigma$-compactness. Without $\sigma$-compactness, the best we can guarantee is that if $X$ and $Y$ are Polish spaces and $Z$ is an arbitrary set, then for any continuous $g : X \times Y\times Z \to \olR$, $f(x) = \inf_{y \in Y} \sup_{z \in Z}g(x,y,z)$ is semi-$\bSigma^1_1$.

\subsection{Logical tameness and o-minimality}
\label{sec:o-min}

%

Another culprit in \cref{thm:polynomial-Vitali} is the \emph{interaction} between quantifying over $\Rb$ and quantifying over $\Nb$. It is of course a standard fact that suprema and infima of countable families of measurable functions are measurable (since measurable sets form a $\sigma$-algebra), but it turns out that if we start from `logically' tame functions, solely quantifying over $\Rb$ can be tame as well.

The earliest named example of this phenomenon is probably the Tarski--Seidenberg theorem, which says that projections of semialgebraic sets are semialgebraic, where a set $X \subseteq \Rb^n$ is \emph{semialgebraic} if it is a finite Boolean combination of polynomial equalities and inequalities (i.e., sets of the form $\{\xbar \in \Rb^n : p(\xbar) = 0\}$ and $\{\xbar \in \Rb^n : p(\xbar) > 0\}$ for polynomial $p$). Semialgebraic sets are locally closed and hence always Borel. 

Since complements of semialgebraic sets are semialgebraic, it follows immediately from the Tarski--Seidenberg theorem that every set $X \subseteq \Rb^n$ that is first-order definable in the reals as an (ordered) field is semialgebraic. In the context of this paper, this has the following consequence.

\begin{fact}[Tarski--Seidenberg]\label{fact:Tarsk-Seidenberg}
  Every function of the form
  \[
    f(x) = \qqq_{v \in \Rb} \qqq_{v \in \Rb} \qqq_{v \in \Rb}\cdots p(x,\dots)
  \]
  (where each $\qqq$ is $\sup$ or $\inf$, each $v$ is a variable, and $p$ is a polynomial with real coefficients) is Borel and therefore universally measurable.
\end{fact}
\begin{proof}
  The $\olR$-valued function $f$ is first-order definable\footnote{We say that a function $f : \Rb^n \to \olR$ is \emph{definable} in some structure on $\Rb$ if the sets $\{\langle \xbar ,y  \rangle \in \Rb^{n+1} : f(\xbar) = y\}$, $\{\xbar \in \Rb^n : f(\xbar) = \infty\}$, and $\{\xbar \in \Rb^n : f(\xbar) = -\infty\}$ are all definable in that structure.} with parameters in $(\Rb,+,\cdot)$. Hence each set of the form $f^{-1}((r,\infty])$ is semialgebraic and thereby Borel.
\end{proof}

This can be extended beyond polynomials to anything first-order definable in $(\Rb,+,\cdot)$, which includes things like $\sqrt{x}$ and rational function (once we fix some reasonable convention for handling normally undefined values). For instance, the function
\[
  f(x) =
  \begin{cases}
    \sqrt{x} & x \geq 0 \\
    0 & x < 0
  \end{cases}
\]
is definable in $(\Rb,+,\cdot)$ by the formula $\varphi(x,y) \equiv (x < 0 \wedge y = 0) \vee (x \geq 0 \wedge y \geq 0 \wedge y \cdot y = x)$.\footnote{Of course the relation $x \leq y$ is definable by the formula $\psi(x,y) \equiv \exists z(x + z \cdot z = y)$.}

%

In the 80s and into the 90s, seminal model-theoretic work of van den Dries, Knight, Pillay, and Steinhorn isolated a general class of structures, the \emph{o-minimal structures}, that admit this kind of automatic topological tameness for definable objects. This extended earlier work of \L ojasiewicz and Hironaka on \emph{semianalytic} and \emph{subanalytic} sets, which exhibit topological tameness analogous to that of semialgebraic sets for certain sets defined in terms of equalities and inequalities of real-analytic functions (see \cite[Ch.~5]{Acquistapace2022}).

\begin{defn}
  A first-order structure $\Rc$ extending $(\Rb,<)$ is \emph{o-minimal} if any definable (with parameters) subset of $\Rb$ is a finite union of intervals (of possibly $0$ or infinite length).
\end{defn}

Note that since semialgebraic subsets of $\Rb$ are always finite unions of intervals, we have that $(\Rb, +, \cdot)$ is o-minimal. This property ends up having profound structural consequences---for instance it implies that every definable set of any number of dimensions has finitely many connected components---and it has found many applications outside of model theory \cite{Binyamini2023,TQFT2024}. This is sometimes regarded as an answer to Grothendieck's challenge to `investigate classes of sets with the tame topological properties of semialgebraic sets' issued in his \emph{Esquisse d’un Programme}. 

\begin{fact}[{Knight, Pillay, Steinhorn \cite{Knight1986}, see also \cite[Ch.~3]{Dries1998}}]
  If $f : \Rb^k \to \olR$ is definable in an o-minimal structure $\Rc$, then for every (open, closed, or half-open) interval $I \subseteq \olR$, $f^{-1}(I)$ is a finite union of connected submanifolds of $\Rb^k$.
\end{fact}

If $f(\xbar,y)$ is definable in an o-minimal structure $\Rc$, then the functions $\inf_{y \in \Rb} f(\xbar,y)$ and $\sup_{y \in \Rb}f(\xbar,y)$ are as well. This means that we are able to generalize \cref{fact:Tarsk-Seidenberg} to any o-minimal structure on $\Rb$.

%

\begin{cor}\label{cor:o-min-meas}
  For any function $g(x,\dots)$ definable in an o-minimal structure $\Rc$, every function of the form
  \[
    f(x) =  \qqq_{v \in \Rb}\qqq_{v \in \Rb}\qqq_{v \in \Rb} \dots g(x,\dots)
  \]
  (where each $\qqq$ is $\sup$ or $\inf$ and each $v$ is a variable) is Borel and therefore universally measurable.
\end{cor}

By Wilkie's theorem \cite{Wilkie1996}, the structure $(\Rb,+, \cdot,e^x)$ is o-minimal, so \cref{cor:o-min-meas} applies to all functions built out of the basic arithmetic operations, $e^x$, $\log x$, and $|x|$, such as $\frac{\log |x|+e^{(y^2-z)/e^x}}{e^{y}+z^3}$.

Not all analytic functions are definable in o-minimal structures. In fact, even $\sin(x)$ is `wild' from this perspective.

\begin{fact}\label{fact:sine-sin}
  A function $f: \Rb^n \to \Rb$ is first-order definable (with parameters) in $(\Rb,+,\cdot,\sin(x))$ if and only if it is $\bSigma^1_n$ for some $n$. 
\end{fact}

\cref{fact:sine-sin} seems to be folklore, but a lot of the machinery needed to prove it is similar to the techniques used in this paper. The key is that $\sin(x)$ allows us to define $\mathbb{Z}$ and thereby define $\mathbb{N}$. This means that \cref{fact:sine-sin} holds for other expansions of $(\Rb,+,\cdot)$ by other seemingly tame functions such as $(\Rb,+,\cdot,\lfloor x \rfloor)$ (where $\lfloor x \rfloor$ is the floor function).

However, there is still some sense in which these familiar functions are tame in terms of quantification and measurability. This was originally studied in the context of semianalytic and subanalytic sets, but for our purposes it will be easier (and slightly more general) to phrase things in terms of o-minimality.

%


\begin{defn}
  A function $f : \Rb^n \to \Rb$ is \emph{locally o-minimal} if there is an o-minimal expansion of the reals in which the restriction of $f$ to $[-k,k]^n$ is definable for each $k$ (although not necessarily uniformly).
\end{defn}

By the main results of \cite{vandenDries1994}, every entire real-analytic function is locally o-minimal. Furthermore, so are many commonly used piecewise continuous functions, such as the floor and ceiling functions and the square, sawtooth, and triangle wave functions.

This regularity assumption only really buys us one additional quantifier, which is non-trivial but only just.

\begin{prop}\label{prop:local-o-min}
  If $h : \Rb^{n + m} \to \Rb$ is locally o-minimal, then $f(\xbar) = \sup_{\ybar \in \Rb^m}h(\xbar,\ybar)$ and $g(\xbar) = \inf_{\ybar \in \Rb^m}h(\xbar,\ybar)$ are Borel and therefore universally measurable.
\end{prop}
\begin{proof}
  We have that $f(\xbar) = \sup_{k \in \Nb} \sup_{\ybar \in [-k,k]^m}h(\xbar,\ybar)$. By local o-minimality, $\sup_{\ybar \in [-k,k]^m}h(\xbar,\ybar)$ is Borel for each $k \in \Nb$, whereby $f(\xbar)$ is Borel. The infimum case follows from the supremum case by considering $-\sup_{\ybar \in \Rb^m}-h(\xbar,\ybar)$.
\end{proof}

Note though that in the case of real-analytic functions, \cref{prop:local-o-min} is not an improvement over \cref{prop:three-quant-insufficient}.


\subsection{Measurability from large cardinals}
\label{sec:meas-large}







It has been an ongoing theme of set-theoretic research that there is an intimate relationship between measurability of definable subsets of $\Rb$ and large cardinals. We've already seen a bit of this in \cref{fact:Shelah-inaccess}. We've also seen (in \cref{fact:Krivine-proj-meas}, for instance) that it can \emph{consistently} be the case that all reasonably definable functions on the reals are measurable, but on a psychological level it would be more comfortable if, for instance, it just were the case that no `explicitly definable' subset of $\Rb^3$ can be a paradoxical decomposition of $S^2$. The impossibility of such things follows from certain beyond-\ZFC\ set-theoretic principles that are commonly considered, such as various large cardinal axioms and \emph{projective determinacy} (a restricted form of the more famous \emph{axiom of determinacy} (\AD) which, unlike \AD, is consistent with the axiom of choice). We don't have enough space in this paper to discuss projective determinacy other than to mention that, like some large cardinal axioms, it has found applications outside of set theory and descriptive set theory, such as in recent work by Avil\'es, Rosendal, Taylor, and Tradacete resolving certain problems in Banach space theory under the assumption of projective determinacy \cite{ARTT2024}.

%

%

\begin{defn}
  A cardinal $\kappa$ is \emph{measurable} if there is a $\kappa$-additive $\{0,1\}$-valued measure $\mu$ on the $\sigma$-algebra of all subsets of $\kappa$ such that $\mu(\kappa)=1$ (where \emph{$\kappa$-additivity} in this context means that for any family $(A_i)_{i \in I}$ of subsets of $\kappa$ with $|I| < \kappa$, $\mu\left( \bigcup_{i \in I}A_i \right) = 1$ if and only if there is an $i \in I$ such that $\mu(A_i) = 1$).
\end{defn}

Measurable cardinals are inaccessible, but this is a bit like saying that the core of the sun is warm. 
%
To give a little perspective, the number of distinct inaccessible cardinals less than a measurable cardinal $\kappa$ is $\kappa$. To give a little bit more perspective, the number of cardinals less than a measurable $\kappa$ with this property is also $\kappa$. Moreover, the number of cardinals $\lambda$ less than $\kappa$ satisfying that ``the number of cardinals $\gamma$ less than $\lambda$ satisfying that `the number of inaccessible cardinals less than $\gamma$ is $\gamma$' is $\lambda$'' is $\kappa$. Furthermore, none of the obvious (even transfinite) iterations of this idea come close to actually characterizing measurability. If $\kappa$ is measurable, then there is a $\kappa$-additive measure $\mu$ on $2^\kappa$ such that
\[
  \mu(\{\alpha < \kappa : \alpha~\text{is an inaccessible cardinal}\}) = 1, 
\]
so there's some sense in which you can `randomly' pick an ordinal less than $\kappa$ and almost surely find an inaccessible cardinal.

Measurable cardinals were defined by Ulam in the investigation of the question of which sets can admit countably additive measures on their full power sets (a generalization of the measure problem of Lebesgue),\footnote{In particular, the existence of a measurable cardinal follows from the assumption that there is a non-trivial countably additive $\{0,1\}$-valued measure on the full power set of some infinite set.} but in some sense the fact that measurable cardinals have implications for measurability in $\Rb$ in particular is a historical coincidence.

\begin{fact}[{Solovay \cite[Thm.~14.3]{Kanamori2003}}]\label{fact:meas-meas}
  If there is a measurable cardinal, then every $\bSigma^1_2$ set of reals is universally measurable.
\end{fact}

\cref{fact:meas-meas} is known to not be sharp (as in weaker large cardinal hypotheses, such as the existence of a Ramsey cardinal, suffice), but measurable cardinals show up in some surprising places in mathematics. The existence of a measurable cardinal is equivalent to the existence of a discrete topological space that is not realcompact (i.e., isomorphic to a closed subspace of $\Rb^I$ for any $I$) and to the existence of an exact functor $F : \Set \to \Set$ that is not naturally isomorphic to the identity (as discovered independently by Trnkov\'a \cite{Trnkova1971} and then later Blass \cite{blass1976exact}). %
It was shown by Prze\'zdziecki in \cite{AdamPrzezdziecki2006} that the \emph{non-}existence of a measurable cardinal is equivalent to the statement that every group is the fundamental group of a compact Hausdorff space. Directly concatenating this with \cref{fact:meas-meas} gives the somewhat absurd result that the existence of a group that is not the fundamental group of any compact Hausdorff space (non-vacuously) implies that the projection of the complement of the projection of any Borel subset of $\Rb^3$ is Lebesgue measurable. 
%
%
%
%
%
In the context of this paper more specifically, the functions in Theorems~\ref{thm:polynomial-Vitali} and \ref{thm:Polynomial-Banach-Tarski} are measurable if there is a measurable cardinal. A corollary of this fact (which is much easier to prove directly and was originally shown by Scott in \cite{SCOTT1961}) is that there cannot be a measurable cardinal in $L$; it is `too thin' to accommodate such an object, in some sense.\footnote{Note though that if there is a measurable cardinal $\kappa$, then the literal ordinal $\kappa$ will still exist in $L$ and will be a (fairly large inaccessible) cardinal there. It just won't be a measurable cardinal in the sense of having a countably additive $\{0,1\}$-valued measure on its power set.} There is however a modified construction, $L[U]$, which has many similar properties to $L$ and can accommodate a measurable cardinal. Silver showed in \cite{Silver1971Meas} that there is a $\lDelta^1_3$ well-ordering of $\Rb$ in $L[U]$,\footnote{See also \cite[Thm.~20.18]{Kanamori2003} for a modern proof of this in terms of \emph{iterable pre-mice}, which are an important tool in studying the fine structure of more sophisticated inner models like $L[U]$.} so we get analogs of Theorems~\ref{thm:polynomial-Vitali} and \ref{thm:Polynomial-Banach-Tarski} for the theory $\ZFC + \text{``}$there exists a measurable cardinal'' at the cost of adding one additional supremum or infimum at the beginning of the expression defining the relevant indicator functions.


  \begin{figure}
    \centering
    \LRFigure
    \caption{$L(\Rb)$ is the smallest inner model containing all of $\Rb$.}%
    %
  \label{fig:LR}
\end{figure}

There are stronger large cardinal assumptions which push the automatic measurability of simply defined functions up the projective hierarchy. The most precise version of these involve the concept of a \emph{Woodin cardinal},\footnote{Woodin cardinals are not usually defined this way, but Woodinness of $\kappa$ is equivalent to $H_\kappa$ satisfying the \emph{weak Vop\v{e}nka's principle} \cite[Def.~6.21]{Adamek1994}, which says that $\Ord^{\mathrm{op}}$ cannot be fully embedded into any locally presentable category \cite{Wilson2021}. In other words, $\kappa$ is a Woodin cardinal if and only if $\kappa^{\mathrm{op}}$ (as a posetal category) cannot be fully embedded into any locally $\kappa$-presentable category of size $\kappa$.} but the required assumption is much weaker than \emph{Vop\v{e}nka's principle}, which is more well known and has already found applications outside of logic. As discussed in \cite[Ch.~6]{Adamek1994}, Vop\v{e}nka's principle is equivalent to many nice statements in the theory of locally presentable categories. For instance, it is equivalent to the statement that a category $C$ is locally presentable if and only if it is co-complete and has a small full subcategory $D$ such that every object of $C$ is a colimit of a diagram in $D$. 



\begin{fact}[{Shelah, Woodin \cite{Shelah1990}}] $ $\label{fact:Woodin-meas}
  \begin{enumerate}
  \item For any $n \in \Nb$, if there is a measurable cardinal that is larger than $n$ Woodin cardinals, then every $\bSigma^1_{n+2}$ set of reals is universally measurable.
  \item\label{second-meas} If there is a measurable cardinal that is larger than infinitely many Woodin cardinals, then every set of reals in $L(\Rb)$ is universally measurable. In particular, this follows from Vop\v{e}nka's principle. %
  \end{enumerate}
\end{fact}
\begin{proof}
  By \cite[Thm.~20.24, Lem.~20.25]{Jech2003}, Vop\v{e}nka's principle implies the existence of a supercompact cardinal, which is known to imply the assumption of \ref{second-meas} (as discussed in \cite{Shelah1990}). The rest follows from \cite[Thm.~32.9]{Kanamori2003}.
\end{proof}

  
$L(\Rb)$ is the smallest inner model containing all of the real numbers (see \cref{fig:LR}). Unlike $L$ though, it doesn't always satisfy the axiom of choice (otherwise it would always have non-measurable sets). I would argue that $L(\Rb)$ encompasses (and probably over-approximates) most mathematicians' intuitive sense of what it means to `define something without the axiom of choice' or in other words to `actually write something down.' Any set of reals or real-valued function that one can prove the existence of without choice will live inside $L(\Rb)$, since it is a model of \ZF. So we have that if Vop\v{e}nka's principle holds, no analogs of Theorems~\ref{thm:polynomial-Vitali} and \ref{thm:Polynomial-Banach-Tarski} are possible. There can be no `explicit' description of a Vitali set or a Banach--Tarski decomposition of the sphere in the presence of sufficiently strong large cardinals.






\subsection{Table of (non-)measurability results}



\cref{tab:meas-tab} summarizes some of the various measurability results we've seen in this section and compares them to the non-measurability results from \cref{sec:construction} (and a couple we'll discuss in a moment). In this table, $\qqq$ represents an instance of either $\inf$ or $\sup$. The asterisks are Kleene stars, so in particular $\sup^\ast_X$ represents an arbitrarily long finite string of suprema over $X$ and $\qqq^\ast_X$ represents an arbitrarily long finite string of (possibly alternating) suprema and infima. The expression $[\sup_\Rb,\qqq_\Nb]^\ast$ represents an arbitrarily long finite string of $\sup_\Rb$'s, $\sup_\Nb$'s, and $\inf_\Nb$'s (again, possibly alternating).  Of course, all of these statements entail the analogous inverted statement, with suprema and infima (and lower and upper) switched, as all of the definability classes mentioned are closed under negation (except lower semi-continuity). Finally, $\mathrm{Con}(\ZFC)$ is the statement that $\ZFC$ is consistent.
%
%

Of course the table is not comprehensive. For instance, by Facts~\ref{fact:Bool} and \ref{fact:meas-meas}, the presence of a measurable cardinal extends the seventh entry of the table to $[\inf_{\Rb},\qqq_\Nb]^\ast\allowbreak[\sup_{\Rb},\qqq_{\Nb}]^\ast\allowbreak\inf_{\Rb,\Nb}^\ast\sup_{\Rb,\Nb}^\ast g$ for continuous $g$ (with analogous extensions for the eighth and ninth entries and in the presence of stronger large cardinal assumptions). There are also more independence results as seen in the third and fourth entries. %
For example, it is possible (although surprisingly tedious\footnote{\textbf{Exercise 1.}  Show that for any open set $U \subseteq \Rb^{n}$, there is an entire real-analytic function $g(\xbar,y)$ such that \(\mathbf{1}_U(\xbar) =  \sup_{y \in \Rb}g(\xbar,y) \) for all $\xbar \in \Rb^n$, where $\mathbf{1}_U(\xbar)$ is the indicator function of $U$.  For extra credit, argue informally that $g(\xbar,y)$ can be computable if $U$ is semicomputable.
  %

 \textbf{Exercise 2.} Show that for any open set $U_0 \subseteq \Rb^k \times \Nb$, there is an open set $U \subseteq \Rb^{k+1}$ such that for any $\xbar \in \Rb^k$, $\langle \xbar,n \rangle \in U_0$ for all $n \in \Nb$ if and only if $\langle \xbar,y \rangle \in U$ for all $y \in \Rb$.

\textbf{Exercise 3.} Use Exercises~1 and 2 to show that for any $G_\delta$ set $A \subseteq \Rb^\ell$, there is a real-analytic function $g(\xbar,y,z)$ such that
  \(
    \mathbf{1}_A(\xbar) = \inf_{y \in \Rb}\sup_{z \in \Rb}g(\xbar,y,z)
  \)
  for all $\xbar \in \Rb^\ell$. Use this and earlier results in this paper to justify the second entry of \cref{tab:meas-tab}.}) to produce a computable total real-analytic function $g(x,y,z,w,t)$ such that it is independent of \ZFC\ whether $f(x) = \inf_{y \in \Rb}\sup_{z \in \Rb}\inf_{w \in \Rb}\sup_{t \in \Rb}g(x,y,z,w,t)$ is Lebesgue measurable, which shows that local o-minimality really doesn't allow us to improve \cref{cor:three-quant-insufficient-1} in general.

  \begin{table}[]
    \begin{adjustbox}{max width=\textwidth}
    \begin{tabular}{rll}
      Function definition & Definability of $g$ & Measurability \\ \hline \hline
      $\inf_{\Rb}\sup_{\Rb}\inf_{\Nb}\sup_{\Nb}^\ast g$                  & Polynomial over $\Zb$        & Independent (\ref{thm:polynomial-Vitali})      \\    %
      $\sup_{\Rb}\inf_{\Rb}\sup_{\Rb}\inf_{\Nb}\sup_{\Nb}^\ast g$                  & Polynomial over $\Rb$        & Implies $\mathrm{Con}(\ZFC)$ (\ref{thm:polynomial-inacc})      \\ %
      $\inf_{\Rb}\sup_{\Rb}\inf_{\Rb}\sup_{\Rb} g$                  & Real-analytic        &   Independent \\ %
      %
      %
      $\inf_{\Rb}\sup_{\Rb}\inf_{\Rb}^\ast\sup_{\Rb}^\ast\inf_{\Rb} g$                  & Trig.~polynomial    & Independent  \\ 
      $\inf^\ast_{\Rb}\sup^\ast_{\Rb}\inf_{\Rb}^\ast\sup_{\Rb}^\ast g$ & Trig.~polynomial & ????? (\ref{quest:trig}) \\ \hline

       %
      %
      %

      $[\sup_{\Rb},\qqq_{\Nb}]^\ast g$                  & Borel        & Measurable (\ref{lem:semi-sigma})      \\ %
      $[\sup_{\Rb},\qqq_{\Nb}]^\ast\inf_{\Rb,\Nb}^\ast\sup_{\Rb,\Nb}^\ast g$                  & Continuous        & Measurable (\ref{lem:semi-sigma}, \ref{cor:three-quant-insufficient-1})      \\ %
                          %
      $[\sup_{\Rb},\qqq_{\Nb}]^\ast\qqq_{\Rb}^\ast g$                  &  O-minimal: Rational,         & Measurable (\ref{lem:semi-sigma}, \ref{cor:o-min-meas})      \\ %
                         &  Exponential, Step, etc.        &       \\  %
                          &  (e.g., $e^{x^2/|y+\mathbf{1}_{[3,4]}(\log z)|}$)        &       \\ %
                          %
                          %
                          %
                          %
     $[\sup_{\Rb},\qqq_{\Nb}]^\ast\inf_{\Rb}^\ast g$  & Locally o-minimal: & Measurable (\ref{lem:semi-sigma}, \ref{prop:local-o-min}) \\ %
        & Real-analytic, Floor, etc. &  \\  
                          & (e.g., $\lceil x^2\cos(e^{y\lfloor z \rfloor})\rceil$) & \\ \hline   %
      %
%
$[\inf_{\Rb},\qqq_{\Nb}]^\ast[\sup_{\Rb},\qqq_{\Nb}]^\ast g$ & Borel &  Measurable cardinal (\ref{fact:meas-meas}) \\
%
      $\qqq_{\Rb,\Nb}^\ast g$ & $L(\Rb)$ & Vop\v{e}nka's principle (\ref{fact:Woodin-meas})
    \end{tabular}
  \end{adjustbox}
  \caption{Measurability of various classes of definable functions.}
    %
    \label{tab:meas-tab}
  \end{table}

Aside from \cref{quest:degree}, the only question I wasn't able to resolve to my own satisfaction was the case of trigonometric polynomials. It is fairly straightforward to show that for any function $g(x_1,\dots,x_n,y,z_1,\dots,z_k)$, 
   \[
     \inf_{y \in \Zb}\sup_{\zbar \in \Zb^k} g(\xbar,y,\zbar) = \inf_{y \in \Rb}\sup_{\beta \in \Rb,\zbar \in \Rb^{k}}\inf_{\gamma \in \Rb} \left[  g(\xbar,y,\zbar) + \beta^2\sin^2(\pi y)   - \gamma^2\sum_{i = 1}^{k} \sin^2(\pi z_i) \right]
   \]
   for any $\xbar \in \Rb^n$. Using a standard argument involving Lagrange's four-square theorem (which allows us to convert quantification over $\Nb$ into quantification over $\Zb$ at the cost of nearly quadrupling the number of variables), this allows us to modify the proof of \cref{thm:polynomial-Vitali} to show that there is a function of the form  $f(x) = \inf_{y \in \Rb}\sup_{ z \in \Rb}\inf_{\nbar \in \Rb^4}\sup_{\beta \in \Rb,\kbar \in \Rb^{280}}\inf_{\gamma \in \Rb} g$ for a trigonometric polynomial $g$ such that whether $f(x)$ is measurable is independent of \ZFC. The fact that something like this should be possible follows from \cref{fact:sine-sin}, but it is unclear if the extra alternation of quantifiers is necessary.

   \begin{quest}\label{quest:trig}
     Does \ZFC\ prove that for every trigonometric polynomial $g$, any function of the form $\inf_\Rb^\ast\sup_\Rb^\ast \inf_\Rb^\ast \sup_\Rb^\ast g$ is Lebesgue measurable?
   \end{quest}

   A positive answer to this would be interesting because it isn't resolved by the general statements \ref{lem:semi-sigma}, \ref{cor:three-quant-insufficient-1}, \ref{cor:o-min-meas}, and \ref{prop:local-o-min}.

   


%


   
%%% Local Variables:
%%% mode: latex
%%% TeX-master: "anything-I-can-write-down"
%%% End: